Author's response to reviews

Title: Clonidine versus captopril for treatment of postpartum very high blood pressure: Study protocol for a randomized controlled trial (CLONCAP)

Authors:

Carlos Noronha-Neto (caconoronha@gmail.com)
Leila Katz (katzleila@gmail.com)
Isabela C Coutinho (isabela@elogica.com.br)
Sabina B Maia (sabinabastosmaia@gmail.com)
Alex Sandro R Souza (alexrolland@uol.com.br)
Melania M Amorim (melania.amorim@gmail.com)

Version: 3 Date: 18 July 2013

Author's response to reviews:

Dear editors, we added the information about the type of study (Protocol) in the title and as we informed previously, both approvation from the Institutional Review Board of the coordinating center (IMIP, Recife, Brazil) and of the National Council for Ethics in Research (CONEP) of the Brazilian Ministry of Health were attached in the first version, as well as the comprovant of the CNPq’s grant. Anyway, we sent them again by e-mail.